% effort-group.tex

\begin{table}

\centering
\begin{tabular}{|c|c|c|c||l|}
\hline
Name1 & Name2 & Name3 & Name4 & Task
\\
email1 & email4 & email4 & email4 & % uniqnames
\\\hline
25 & 25 & 25 & 25 & Example % delete this row
\\
? & ? & ? & ? & leadership
\\
? & ? & ? & ? & planning/design
\\
? & ? & ? & ? & programming
\\
? & ? & ? & ? & testing
\\
? & ? & ? & ? & analysis
\\
? & ? & ? & ? & proposal
\\
? & ? & ? & ? & progress report
\\
? & ? & ? & ? & presentation
\\
? & ? & ? & ? & report
\\
? & ? & ? & ? & code repo
\\
? & ? & ? & ? & (add more rows as needed) 
\\\hline
\end{tabular}

\caption{
Group Project Effort
}
\end{table}


(You need not include the following instructions in your report.)

Purpose
\\
To have each group evaluate member contributions.

Remarks
\\
Some variation in individual effort by group members is expected.
As stated in the project guidelines, members of a group will not
necessarily be given the same grade for their project.

Instructions
\\
Complete the table above.
Each row of values should sum to 100\% (or should be blank if inapplicable).
Add rows if needed for other contributions.
Feel free to add any other relevant comments.

The Honor Code applies here.
It is very unlikely that every team member
did exactly equal parts of all components.
